Author’s response to reviews

Title: Development of breast cancer mortality, considering the implementation of mammography screening programs - a comparison of western European countries

Authors:

Yukio Iwamoto (yukio.iwamoto@uni-heidelberg.de)

Simone Kaucher (simone.kaucher@uni-heidelberg.de)

Eva Lorenz (eva.lorenz@bnitm.de)

Till Bärnighausen (till.baernighausen@uni-heidelberg.de)

Volker Winkler (v.winkler@uni-heidelberg.de)

Version: 3 Date: 12 Jun 2019

Author’s response to reviews:

Assistant Editor Comments:

1. Ethics

-- Please confirm in the section 'Ethics Approval and Consent to Participate' whether informed consent, written or verbal, was obtained from all participants and clearly state this in your manuscript. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

-- If ethics is not required because data is publically available secondary data, please clearly state this.

→ We added this information on page 11

2. Data Permisson
Please state whether the data obtained for your study was publically available under 'availability of data and materials'. If data is not publically available please state any permissions obtained to access the data.

→ We added this information on page 11

3. Authors Contributions

→ Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

→ We added this information on page 11

4. Abbreviations

→ If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided just before the Declarations.

→ We added this information on page 11

5. Formatting

Please remove the point by point and cover letter as this is not required at this stage in the editorial process.

→ done

6. Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional
files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

BMC Public Health operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

→ done

Reviewer reports:

Robert Smith (Reviewer 3): Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format.

→ noted